Author's response to reviews

Title: Lower hamstring extensibility in men compared to women is explained by differences in stretch tolerance

Authors:

Paul W.M Marshall (p.marshall@uws.edu.au)
Jason C Siegler (j.siegler@uws.edu.au)

Version: 2
Date: 20 June 2014

Author's response to reviews:

Response attached.